Massachusetts Institute of Technology - Physics Department

Physics - 8.02 Information on the Final Exam May 10, 2002.

When & Where?

The final Exam will be held on Tuesday, May 21 from 9 AM to 12 noon in the Johnson Athletics Center.

Material Covered

This exam will cover:

(i) all material discussed in lectures throughout the course (ii) all associated reading (iii) all homework assignments.

This is a closed book exam. We will provide you with many equations. You may not use a calculator, and you are not allowed to bring any notes.

There is no make-up Final.

We cannot excuse you from taking the Final Exam, only a dean in Counseling and Support Services can do that. Please keep in mind that if you don’t take the Final (for whatever reason), you will receive a letter grade “o”.

Grading Policy See the General Information, handed out during the first lecture. You can find it on the 8.02 Home Page under ”general information”.

Practice Tests Several tests (with solutions for some) given in previous years, which contain topics that have a large overlap with ours, can be accessed on our website under “previous years”.

Tutoring Schedule See the 8.02 Home Page under ”tutoring”.

Good Luck! Walter Lewin